Author's response to reviews

Title: Improving Motivation and Retention of Health Professionals in Developing Countries: A Systematic Review

Authors:

Mischa Willis (willismm@tcd.ie)
Posy Bidwell (bidwellp@tcd.ie)
Stephen Thomas (steve.thomas@tcd.ie)
Laura Wyness (LWyness@whc.ie)
Duane Blaauw (duane.blaauw@nhls.ac.za)
Prudence Ditlopo (prudence.ditlopo@nhls.ac.za)

Version: 5 Date: 30 October 2008

Author's response to reviews:

Dear Sir or Madam,

In terms of the second review of the draft, no changes were asked for by any of the reviewers. We have though updated the list of author names and noted the corresponding author.

Kind Regards

Dr Steve Thomas